Author’s response to reviews

Title: Does women's caste make a significant contribution to adolescent pregnancy in Nepal? : A study of Dalit and non-Dalit adolescents and young adults in Rupandehi district.

Authors:

Hridaya Devkota (hridayadevkota@hotmail.com)

Andrew Clarke (andrew@kidasha.org)

Shanti Shrish (shanti_shrish@hotmail.com)

Dharma Bhatta (dnbhatta@yahoo.com)

Version: 2 Date: 22 Dec 2017

Author’s response to reviews:

Response to Editor

1. Removed previous additional files.

2. Tables cited and formatted following the guidelines

3. Included copies of questionnaire (both English and Nepali)